Author's response to reviews

Title: Evaluation of the Kinase Domain of c-KIT in Canine Cutaneous Mast Cell Tumors

Authors:

Joshua D Webster (webste28@cvm.msu.edu)
Matti Kiupel (kiupel@dcpah.msu.edu)
Vilma Yuzbasiyan-Gurkan (yuzbasiyan@cvm.msu.edu)

Version: 4 Date: 22 March 2006

Author's response to reviews:

Dear Robin Stewart,

In order to address the request for formatting changes to the manuscript entitled "Evaluation of the Kinase Domain of c-KIT in Canine Cutaneous Mast Cell Tumors" (MS: [4551309108897823]) we have verified that the authors, title, and abstract are identical to those listed in the submission form, with no extraneous capitalization of the title; we have added complete addresses for each author's affiliations; we have removed page 21, which was extraneous; and we have added reformatted figure 4. We have also reviewed the manuscript for additional typographical errors. We hope that these changes will sufficiently address any formatting concerns. If there are any additional concerns please let us know.

Thank you,

Joshua Webster